Author’s response to reviews

Title: The feasibility study of transnasopharyngeal esophagus echocardiography in the ultrasonic diagnosis

Authors:

Wang Xiaofeng (wangxiaofeng966@163.com)
Fang Nie (fang-nie@163.com)
Na Ye (13649312039@163.com)
Xuehui Liu (208498194@qq.com)
Shaoqing Yang (yangshaoqing999@163.com)

Version: 1 Date: 08 Feb 2019

Author’s response to reviews:

Reviewer #1: In the future, we will choose patients suitable for TNPEE in our daily work and expand the number of patients examined.

Reviewer #2: Written English has been widely edited. The real quality data of TNPEE will be further supplemented in future work. Direct comparison of TNPEE with TTE and/or TEE in the same patient may be difficult. If patients need follow-up review, different examination methods can be selected. Population ages range from 44 to 58; indeed, they are not representative. In the future, we will gradually expand the age range in our daily work. I also think TNPEE contraindication may be an important limitation.